

To Reviewers 1 and 2,

Thank you very much for your helpful feedback regarding our manuscript submission, "A Multi-institutional Retrospective Analysis of FOLFIRI in Patients with Advanced Biliary Tract Cancers."

We have updated the manuscript to reflect stylistic changes. Thank you again for the recommendations and feedback. We appreciate the time and effort you took to help us improve our manuscript.

Sincerely,

Jonathan Mizrahi, MD